Author’s response to reviews

Title: Foodborne botulism due to ingestion of home canned green beans

Authors:
Dorothea Hellmich (dorohell@web.de)
Katja Wartenberg (katja.wartenberg@medizin.uni-halle.de)
Stephan Zierz (stephan.zierz@uk-halle.de)
Tobias Mueller (tobias.mueller@medizin.uni-halle.de)

Version: 2 Date: 11 Nov 2017

Author’s response to reviews:
Dear Dr. Nagpal,

We have carried out the additional clarification (Editors report from 24 Oct 2017):

Regarding the first point we changed the manuscript.
Regarding the suggested discussion of the diagnostic value of the electrophysiological studies this has been included in the discussion.

With Kind regards,
Dorothea Hellmich, MD